

Reconciling the Carbon Balance of Northern Sweden Through Integration of Observations and Modelling

Anusha Sathyanadh, Guillaume Monteil, Marko Scholze, Anne Klosterhalfen, Hjalmar Laudon, Zhendong Wu, Christoph Gerbig, Wouter Peters, Vladislav Bastrikov, Mats B. Nilsson, et al.

▶ To cite this version:

Anusha Sathyanadh, Guillaume Monteil, Marko Scholze, Anne Klosterhalfen, Hjalmar Laudon, et al.. Reconciling the Carbon Balance of Northern Sweden Through Integration of Observations and Modelling. Journal of Geophysical Research: Atmospheres, 2021, 126 (23), 10.1029/2021JD035185. hal-03487722

HAL Id: hal-03487722 https://hal.science/hal-03487722

Submitted on 21 Feb2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



JGR Atmospheres

RESEARCH ARTICLE

10.1029/2021JD035185

Special Section:

Understanding carbon-climate feedbacks

Key Points:

- Northern Sweden region was a carbon sink in the years 2016–2018 according to vegetation model estimates and observations
- Both concentration and flux measurements suggest 2018 drought had limited impact on the Nordic carbon balance
- Empirical measurements can improve vegetation models by integrating them into model simulations at site and regional scale
- However, annual net ecosystem exchange values generated by the models varied between -275 and -25 gC m⁻² year⁻¹

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to: A. Sathyanadh, anusha.sathyanadh@gmail.com

Citation:

Sathyanadh, A., Monteil, G., Scholze, M., Klosterhalfen, A., Laudon, H., et al. (2021). Reconciling the carbon balance of Northern Sweden through integration of observations and modelling. *Journal* of *Geophysical Research: Atmospheres 126*, e2021JD035185. https://doi. org/10.1029/2021JD035185

Received 3 MAY 2021 Accepted 3 NOV 2021

Author Contributions:

Conceptualization: Anusha Sathyanadh, Marko Scholze, Mats B. Nilsson, Matthias Peichl Data curation: Anusha Sathyanadh, Anne Klosterhalfen

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Reconciling the Carbon Balance of Northern Sweden Through Integration of Observations and Modelling

Anusha Sathyanadh¹ ^(D), Guillaume Monteil² ^(D), Marko Scholze² ^(D), Anne Klosterhalfen¹ ^(D), Hjalmar Laudon¹ ^(D), Zhendong Wu², Christoph Gerbig³ ^(D), Wouter Peters⁴ ^(D), Vladislav Bastrikov⁵ ^(D), Mats B. Nilsson¹ ^(D), and Matthias Peichl¹ ^(D)

¹Department of Forest Ecology and Management, Swedish University of Agricultural Sciences, Umeå, Sweden, ²Department of Physical Geography and Ecosystem Science, Lund University, Lund, Sweden, ³Max Planck Institute for Biogeochemistry, Jena, Germany, ⁴Department of Meteorology and Air Quality, Environmental Sciences Group, Wageningen University, Wageningen, The Netherlands, ⁵Laboratoire des Sciences du Climat et de l'Environnement, Institut Pierre-Simon Laplace, CEA-CNRS-UVSQ, Gif-sur-Yvette, France

Abstract The boreal biome plays an important role in the global carbon cycle. However, current estimates of its sink-source strength and responses to changes in climate are primarily derived from models and thus remain uncertain. A major challenge is the validation of these models at a regional scale since empirical flux estimates are typically confined to ecosystem or continental scales. The Integrated Carbon Observation System (ICOS)-Svartberget atmospheric station (SVB) provides observations including tall tower eddy covariance (EC) and atmospheric concentration measurements that can contribute to such validation in Northern Sweden. Thus, the overall aim of this study was to quantify the carbon balance in Northern Sweden region by integrating landatmosphere fluxes and atmospheric carbon dioxide (CO_2) concentrations. There were three specific objectives. First, to compare flux estimates from four models (VPRM, LPJ-GUESS, ORCHIDEE, and SiBCASA) to tall tower EC measurements at SVB during the years 2016–2018. Second to assess the fluxes' impact on atmospheric CO_2 concentrations using a regional transport model. Third, to assess the impact of the drought in 2018. The comparison of estimated concentrations with ICOS observations helped the evaluation of the models' regional scale performance. Both the simulations and observations indicate there were similar reductions in the net CO₂ uptake during drought. All the models (except for SiBCASA) and observations indicated the region was a net carbon sink during the 3-year study period. Our study highlights a need to improve vegetation models through comparisons with empirical data and demonstrate the ICOS network's potential utility for constraining CO₂ fluxes in the region.

Plain Language Summary As the largest global carbon sink, the boreal region can play an important role in climate mitigation strategies. Thus, it is important to estimate the region's sink strength properly. However, due to limitations in measurement techniques at the regional scale, the sink strength estimates are often obtained from models, which require robust validation. Therefore, we combined flux and atmospheric concentration data from one of the stations in Northern Europe to validate estimates generated using four vegetation models. We also used a transport model to connect the fluxes to atmospheric concentration dynamics, then compared the modeled CO_2 concentration patterns with observations to provide a regional scale validation of the vegetation models. These observed and modeled fluxes and concentrations are also used to investigate the impact of the 2018 drought over Nordic vegetation. We find that the 2018 European drought had only limited impact on the carbon fluxes over Northern Sweden. Both the modeling and measurements show that the study region was a carbon sink during the study period (2016–2018), with some variation in estimates from the four models.

1. Introduction

The boreal region plays a major role in the global carbon cycle as it covers 6%–13% of earth's land area and stores ca. 30% of the global terrestrial carbon (Alekseychik et al., 2017; Brandt et al., 2013; Kasischke, 2000). Moreover, boreal forest ecosystems commonly act as a net sink of carbon dioxide (CO₂) and thus strongly contribute to stabilization of the global climate system (Malmsheimer et al., 2011; Pan et al., 2011; Xu et al., 2020). However, although the region is the largest biome in the world with the most widespread type of vegetation in the Northern hemisphere (Gauthier et al., 2015), its annual carbon budget is highly uncertain and interannually



Formal analysis: Anusha Sathyanadh, Guillaume Monteil

Funding acquisition: Matthias Peichl Investigation: Anusha Sathyanadh, Guillaume Monteil, Marko Scholze, Matthias Peichl

Methodology: Anusha Sathyanadh, Guillaume Monteil, Marko Scholze, Mats B. Nilsson

Project Administration: Matthias Peichl Resources: Marko Scholze, Zhendong Wu, Christoph Gerbig, Wouter Peters, Vladislav Bastrikov, Matthias Peichl Software: Anusha Sathyanadh, Guillaume Monteil, Anne Klosterhalfen Supervision: Guillaume Monteil, Marko Scholze, Hjalmar Laudon, Mats B. Nilsson, Matthias Peichl Validation: Anusha Sathyanadh, Guillaume Monteil Visualization: Anusha Sathyanadh Writing – original draft: Anusha Sathyanadh

Writing – review & editing: Anusha Sathyanadh, Guillaume Monteil, Marko Scholze, Anne Klosterhalfen, Hjalmar Laudon, Zhendong Wu, Christoph Gerbig, Wouter Peters, Vladislav Bastrikov, Mats B. Nilsson, Matthias Peichl variable (Smith et al., 2020). There are urgent needs to address these uncertainties and rigorously characterize the variations, for several reasons. First, it is well known that the increasing atmospheric CO_2 concentration (Mittler & Blumwald, 2010; Pachauri et al., 2014) has resulted in global warming and unprecedented climatic changes (Barber et al., 2000; Serreze et al., 2000). Moreover, "Arctic amplification" has contributed to dramatic melting of Arctic Sea ice and spring snow cover (Cohen et al., 2014), but clarification of the extent to which these climatic changes will modify the boreal region's role in the global carbon cycle is needed (Bradshaw & Warkentin, 2015; Pan et al., 2011; Tagesson et al., 2020). Second, the Nordic countries are striving to develop more sustainable low-carbon societies (Urban & Nordensvärd, 2018), and Sweden in particular aims to become the first "fossil (fuel)-free welfare nation" in the world, with zero net emissions of greenhouse gases by 2045 (Xylia & Silveira, 2016). Efforts to achieve this ambitious goal would be helped by thoroughly verified negative emissions that could offset some anthropogenic greenhouse gas emissions (Fauré et al., 2019; Rodriguez et al., 2020). Thus, it is essential to obtain robust and precise estimates of the national carbon balance including the sink strength of the boreal forest region.

Climate change and the associated temperature and precipitation trends trigger both positive and negative carbon cycling feedbacks (Braswell et al., 1997; McGuire et al., 2009). Understanding of biosphere-climate feedback at a regional scale is necessary to evaluate the boreal forest ecosystems' potential to support efforts in mitigating global climate change (Kondo et al., 2020). However, current understanding of these processes at a regional scale primarily relies on terrestrial biosphere models (Cramer et al., 2001; Friend et al., 2007; Haas et al., 2013). Hence, comprehensive evaluations of these models with observations covering multiple spatiotemporal scales is required to assess the robustness of their estimates of biospheric responses to changes in meteorological variables (Baldocchi et al., 2001; Jung et al., 2007; Mahecha et al., 2010; Moorcroft, 2006; Siqueira et al., 2006).

A key variable to consider is Net Ecosystem CO_2 Exchange (NEE). This is the net flux resulting from the major vertical CO_2 fluxes between an ecosystem and the atmosphere (Baldocchi, 2003) and a primary measure of the ecosystem's carbon sink-source strength (Kramer et al., 2002). It is generally determined by subtracting fluxes into the atmosphere generated by soil and plant respiration from fluxes in the other direction generated by photosynthesis. Other components (such as lateral transfers of carbon in and out of the ecosystem; emission of volatile organic carbon, methane, and carbon monoxide; release of soot and CO_2 from fire; and fluxes from terrestrial inland water bodies) also contribute to the "Net Ecosystem Carbon Balance" (NECB) (Chapin et al., 2006). Available models account for these flux components in various ways. Here, the term NEE refers to the net ecosystem land-atmosphere carbon flux from atmosphere to biosphere) and ecosystem respiration (Reco, regarded as the total carbon flux from atmosphere). Discrepancies in estimates of the NEE derived using different methods limit the understanding of interactions between Nordic ecosystems and atmospheric CO_2 , as well as these ecosystems' likely responses to future climate change and associated meteorological events.

Current estimates of the various flux components are generated by one of three approaches: ground-based field measurements (Riederer et al., 2014; Ramonet et al., 2020), bottom-up (vegetation/ecosystem models) modeling (Peng, 2000; Bastos et al., 2020), and top-down (atmospheric inversion) modeling (Gurney et al., 2002; Monteil et al., 2020). In field studies, eddy covariance (EC) techniques are often used to measure exchange fluxes directly and estimate seasonal dynamics and interannual variations of NEE, GPP, and Reco (Law et al., 2000, 2002; Stoy et al., 2006; Schmidt et al., 2012). The establishment of regional and global networks of ground-based micro-meteorological EC flux towers has provided continuous and instantaneous data on the atmosphere-biosphere exchange of water, energy, and carbon via fluxes to and from all types of ecosystems (Baldocchi et al., 2001; Papale et al., 2006). When sensors are mounted on tall towers (at appropriate heights), EC measurements can provide indications of landscape-scale fluxes spanning several tens of km² (Chi et al., 2019, 2020).

In addition, terrestrial vegetation/ecosystem models have become powerful tools for filling measurement gaps, extrapolating observed trends to responses in anticipated future conditions, predicting climate change impacts, understanding processes in broader contexts, and retrospective analyses of ecosystem responses to previous climatic changes (Amthor et al., 2001). They can be used to estimate the growth of vegetation and changes in carbon storage between different carbon pools (leaves, wood, roots, and soils) at various spatial and temporal scales (Pan et al., 2014). Ecosystem models vary in complexity of structures, parameters, and processes that control photosynthetic carbon uptake and respiratory release. Thus, the models' estimates of terrestrial primary production depend on the underlying assumptions, their structure, and formulation. They range from purely empirical models





Figure 1. Map showing the regional domain used in the transport simulations, with blue filled circles marking ICOS atmospheric observation sites. Geographical regions in the south, central, and north are color-coded as follows: peach-southern Sweden and Denmark; green-Southern Norway; blue-Central Sweden; magenta-Northern Boreal Sweden; purple-Central Norway; brown-Arctic Sweden, Northern Norway and Northern Finland; and gray-Southern and Central Finland.

to process-based ecosystem models that can be used to investigate responses of ecosystem processes to both environmental and anthropogenic factors at regional and global scales (Melillo et al., 1993; Tian et al., 2010). While these bottom-up models can provide detailed estimates of the fluxes and characterization of their spatial and temporal variability that can be applied in future climate simulations, large uncertainties may be propagated into the cumulative fluxes (Sitch et al., 2015).

Carbon fluxes can be highly variable even within a single biome (Schulze et al., 1999), so validating the model's estimate at one location does not necessarily guarantee its validity at a large scale. Since the land-atmosphere carbon fluxes clearly affect the atmospheric CO_2 content, they can also be estimated from observations of atmospheric CO_2 concentration. For this, an atmospheric transport model that computes the concentrations as functions of fluxes is needed. Because of atmospheric mixing, fluxes at one location influence CO_2 concentrations over a large area and conversely a single CO_2 observation is sensitive to flux changes over a large area. However, estimates from these approaches sometimes significantly differ, and the discrepancies are more pronounced for boreal forests (where carbon fluxes are relatively small) than in many other ecosystems (Campioli et al., 2016). Therefore, to improve understanding of the boreal biosphere's role in the climate system, it is essential to reconcile the carbon flux estimates from ecosystem to global scales (Baldocchi et al., 2001; Ballantyne et al., 2021; Running et al., 1999).

Valuable data for such purposes are provided by the ICOS SVB (Integrated Carbon Observation System- Svartberget) atmospheric site, including tall tower EC and atmospheric concentration measurements that collectively facilitate novel model-data fusion studies covering a homogenous boreal landscape that is very similar to landscapes in many parts of Northern Sweden (Lindström et al., 2002; Laudon et al., 2021). Another five ICOS sites across the Nordic region provide CO_2 concentration measurements that can be compared to modeled concentrations from an atmospheric transport model, driven by an estimate of the regional carbon fluxes. In this Nordic region, there was a severe drought in 2018, which was even stronger in some places than the drought that occurred in central Europe in 2003 (Lindroth et al., 2020). The aim of this study is to reconcile the carbon balance of Northern Sweden by integrating fluxes from models and observations over Scandinavia at site to regional scales. There were three specific objectives. First, to compare model estimates of the fluxes and concentrations with site-scale observations during the years 2016–2018. Second, to determine how well the atmospheric observation network can constrain the fluxes in the Nordic region. Third, to compare modeled and observed carbon fluxes and atmospheric concentrations during the drought in 2018 (one of the three warmest years on record for Europe; WMO, 2019, Figure S1 in Supporting Information S1) at the ICOS SVB site and the Northern Sweden region.

Table 1

Sites of CO₂ Flux and Concentration Observations Used in This Study

Product type	Station name	Location (Coordinates)	Sampling heights used in the study (m a.g.l.)	Data used in the study	Provider
Flux observation	Svartberget (SVB)	64.25°N, 19.77°E	70 m	2016-01 to 2018-12	SLU
In-situ concentration Observation	Svartberget (SVB)	64.25°N, 19.77°E	70 m	2017-05 to 2018-12	SLU (via ICOS)
	Pallas-Sammaltunturi (PAL)	67.97°N, 24.11°E	12 m	2017-08 to 2018-12	FMI (via ICOS)
	Puijo (PUI)	62.90°N, 27.65°E	84 m	2017-01 to 2018-12	FMI & UEF (via ICOS)
	Hyltemossa (HTM)	56.09°N, 13.41°E	150 m	2017-01 to 2018-12	LU (via ICOS)
	Norunda (NOR)	60.08°N, 17.47°E	100 m	2017-01 to 2018-12	LU (via ICOS)
	Hyytiälä (SMR)	61.84°N, 24.29°E	125 m	2017-01 to 2018-12	UH (via ICOS)

Note. FMI, Finnish Meteorological Institute; UEF, University of Eastern Finland.

2. Models and Data

We used flux estimates from four gridded vegetation models. Two are dynamic global vegetation (DGVM) models: the ORCHIDEE (Organizing Carbon and Hydrology In Dynamic Ecosystems) and LPJ-GUESS (Lund-Potsdam-Jena General Ecosystem Simulator) models. The others are diagnostic models: the VPRM (Vegetation Photosynthesis and Respiration) and SiBCASA (Simple Biosphere/Carnegie-Ames-Stanford Approach) models. We assessed their performance by comparing results they yielded with empirical data at two spatial scales, as follows.

In the first approach, we compared modeled and observed fluxes of NEE and its components Reco and GPP at site level. The selected modeled fluxes are those generated for the nearest model grid cell to the SVB observation site (64.25°N, 19.77°E) during the study period. The observed fluxes are from the tall tower EC data set, which are typically representative of the net flux at landscape scale (Chi et al., 2019). These direct comparisons reveal the vegetation models' ability to reproduce observed fluxes at site scale. Fluxes yielded by the four models for the Northern Sweden region are also compared.

In the second approach, we compared atmospheric concentration measurements at a large scale, which requires forward simulations from an atmospheric transport model. In this study, we used the FLEXPART 10.4 (Flexible Particle dispersion model) Lagrangian transport model (Pisso et al., 2019) to compute flux transport in a regional domain ranging from 0°W, 54°N to 20°E, 70°N (Figure 1). In addition, background concentrations (boundary condition) were taken from a global TM5 inversion, following the LUMIA (Lund University Modular Inversion Algorithm) setup described by Monteil and Scholze, 2021. For the sake of readability, we refer to this transport model setup as LUMIA in the rest of the paper. LUMIA computes the CO₂ concentrations at the six ICOS sites in the domain (Figure 1) using estimates of the surface CO₂ fluxes (NEE and anthropogenic emissions), gridded on a $0.5^{\circ} \times 0.5^{\circ}$, 3-hourly resolution (fluxes were regridded by aggregation or re-binning when necessary). Monthly anthropogenic emissions are based on a pre-release of the EDGAR4.3 inventory (Janssens-Maenhout et al., 2019) for the base year 2010 (doi: 10.18160/Y9QV-S113). At all six sites, continuous CO₂ observations are made at several vertical levels. Only the upper level at each site was computed in LUMIA, as these are where the accuracy of the transport model is expected to be the highest.

The in-situ observations of fluxes and concentrations are described in detail in Section 2.1. Table 1 provides a summary of the observations of fluxes and concentrations and Table 2 provides the model estimates, their sources, original resolution, and geographical position of the sites. Section 2.2 details the structure and functioning of the models and differences between them, while Section 2.3 provides a detailed description of the LUMIA atmospheric transport model. The study period covers the years 2016–2018 (inclusive).

2.1. Observations

2.1.1. Landscape-Scale CO, Flux Estimates From Tall Tower Eddy Covariance Measurements

Tall tower EC flux data are collected at the combined ecosystem-atmosphere ICOS-Svartberget tower (SVB, https://data.icos-cp.eu/portal/) located in the Swedish Agricultural University's Experimental Forests near



Details of Model Simulations (including Boin Gridded and Sile-Specific LPJG Simulations) Used in the Study					
Product type	Product	Spatial resolution	Provider		
Gridded simulation	LPJ-GUESS	$0.5^\circ,0.5^\circ$ for the domain 0°E, 54°N to 30°E, 70°N (shown in Figure 1)	LU, Sweden		
	VPRM		MPI, Germany		
	SiBCASA		WUR, Netherlands		
	ORCHIDEE		LSCE, France		
Site simulation	LPJG_PNV	64.25°N, 19.77°E	LU, Sweden		
	LPJG_Conif				
LUMIA transport simulations	FLEXPART regional transport model	$0.5^{\circ} \times 0.5^{\circ}$	LU, Sweden		

Table 2

Vindeln, in the Krycklan catchment, Northern Sweden (Chi et al., 2020). In addition to the ICOS ecosystem-level EC system (mounted at 34.5 m), an additional EC system was installed on the same 150 m tall tower to measure CO₂, water vapor, and energy exchange between the boreal landscape and atmosphere. This EC system was mounted at 70 m height above the ground from March 2016 until October 2017, then lowered to a measurement height of 60 m. These tall tower EC measurements are typically representative of the net fluxes at landscape-scale (covering ca. 3–4 km around the tower, giving a \sim 25–50 km² footprint for the 60 m measurement level). Hereafter, we refer to data provided by this landscape-level EC system as "EC data". The EC measurement set-up consists of a CSAT3 3D sonic anemometer and EC155 closed-path gas analyzer (Campbell Scientific Inc., USA). Further information about the data processing, corrections, quality checks and controls, and final NEE estimations are reported in detail by Chi et al. (2019). The set of NEE measurements used here was gap-filled and source-partitioned into the two flux components, GPP and Reco, using the R-package REddyProc (v3.6.3, Wutzler et al., 2018). All data processing steps closely followed the ICOS protocol, but it should be noted that the EC system is not part of the ICOS network.

2.1.2. Atmospheric CO, Concentration Measurements

We collected continuous in situ observations of atmospheric CO₂ concentration from ICOS tall towers (ICOS, 2019) (https://meta.icos-cp.eu/collections/4H3RS8YtXIt_WTcjrSSkaQ-O; doi: 10.18160/RHKC-VP22). These included observations from six sites in our regional study domain, for 2017 and 2018 with data availability changing from site to site within that period. Data for 2016 are available at some stations that are not considered for maintaining the consistency across all stations. Most of these high frequency CO₂ concentration measurements were recorded at several vertical levels up to 150 m using tall towers that are now included in the European ICOS atmospheric network. We selected the observations at the highest level from each site. All ICOS sites in the domain where concentration measurements used in the study were marked in Figure 1. Hereafter the term "observations" refers to the measured CO₂ concentrations, unless otherwise specified.

2.2. Ecosystem and Diagnostic Models

2.2.1. LPJ-GUESS (Lund-Potsdam-Jena General Ecosystem Simulator)

LPJ-GUESS (Smith et al., 2014; Smith, 2001) is a process-based terrestrial vegetation model that can predict structural, compositional, and functional properties (including biomass and height) of native ecosystems at landscape, regional, and global scales. It is a dynamic global vegetation model that includes detailed description of the biochemistry (unlike diagnostic models) of the focal vegetation and representations of the energy absorption, carbon, and nitrogen cycling. The current version of LPJ-GUESS includes an interactive plant and soil nitrogen cycle and associated constraints on plant production and ecosystem carbon balance. Photosynthesis is modeled with a simplified Farquhar scheme (Collatz et al., 1991) and vegetation growth is simulated dynamically in a so-called cohort-mode representing different tree age classes in a grid cell. LPJ-GUESS simulations are driven by the 6-hourly Climatic Research Unit Japanese Reanalysis (CRU-JRA) meteorological data set (Weedon et al., 2014) as driving data. Fast (plant physiology) processes are implemented on the time step of the meteorological input (air temperature, precipitation, and radiation) data and slow processes (vegetation dynamics) are implemented annually. Atmospheric CO₂ concentrations are input on a monthly basis and the model applies a static soil type map bucket model for soil hydrology. More details are provided by Smith et al. (2014). We used





Figure 2. Comparison of weekly sums of: (a) net ecosystem carbon exchange (NEE), (b) gross primary production (GPP), and (c) ecosystem respiration (Reco) estimated with four vegetation models (LPJ-GUESS, VPRM, SiBCASA, and ORCHIDEE) and landscape-level eddy covariance measurements (EC data) at the Svartberget (SVB) site from 2016 to 2018.

two site-specific NEE simulations generated by the LPJ-GUESS model forced with local site meteorological data (surface air temperature, precipitation and incoming shortwave radiation measured at the site for the years 2015–2018) in the time series analysis at site scale (Figure 2). For the other analysis presented here, we used flux estimates from the LPJ-GUESS gridded simulations. The site-level runs were carried out without site-specific calibration, using inputs of nitrogen deposition and soil properties from global datasets (Smith et al., 2014). One (denoted by LPJG_PNV) involved use of the PNV (Potential Natural Vegetation) vegetation type that establishes under the model's dynamic vegetation processes (at SVB: boreal evergreen coniferous trees, understory grass and some boreal summer-green trees). In the other (designated LPJG_Conif), the plant functional type that could establish was restricted to match the SVB site vegetation cover (boreal evergreen coniferous trees and understory grass), but neither site management history nor disturbances were considered. Finally, no wetland features were activated in the site simulations.

2.2.2. ORCHIDEE (Organizing Carbon and Hydrology in Dynamic Ecosystems)

ORCHIDEE is a global process-based terrestrial biosphere model developed as a land surface component of Institut Pierre-Simon Laplace (IPSL)-CM5 earth system model (Krinner et al., 2005 and http://www.ipsl.jussieu. fr/~ssipsl/). It computes carbon, water, and energy fluxes between the land surface and atmosphere, as well as within the soil-plant continuum. The model computes GPP with CO_2 assimilation based on functions presented by Farquhar et al., (1980) for C3 plants. Land cover changes (including deforestation, regrowth, and cropland dynamics) were prescribed using annual land cover maps derived from the harmonized land use data set (Hurtt et al., 2011) in combination with the European Space Agency- Climate Change Initiative (ESA-CCI) land cover products. The carbon module estimates: growth and maintenance respiration, which sum up to autotrophic



respiration (Ruimy et al., 1996); Net Primary Production (NPP); heterotrophic respiration (Rh); and NEE (NEE = NPP-Rh) at 3 hourly temporal and 0.125° spatial scale using the CRU-HARMONIE meteorological input data (Petrescu et al., 2021). ORCHIDEE has been shown to simulate responses of natural ecosystems in Europe to climatic anomalies, such as the 2003 heat wave, reasonably well (Ciais et al., 2005).

2.2.3. VPRM (Vegetation Photosynthesis and Respiration Model)

VPRM (Mahadevan et al., 2008), a diagnostic model, is a satellite-based assimilation scheme that estimates hourly NEE values based on a light-use efficiency approach and temperature-dependent estimates of ecosystem respiration using high-resolution data. It has an extremely simple mathematical structure, with minimal parameters. VPRM uses ECMWF (European Centre for Medium-Range Weather Forecast) operational meteorological data for radiation and temperature inputs, the SYNMAP land cover classification scheme (Jung et al., 2006), together with EVI (Enhanced Vegetation Index) and LSWI (Land Surface Water Index) values derived from Moderate Resolution Imaging Spectroradiometer (MODIS) reflectance data. Model parameters were optimized for Europe using EC measurements collected during 2007 from 47 sites (Kountouris et al., 2015). The VPRM simulations used here has been produced at 0.5° spatial and hourly temporal resolution. VPRM also provides partitioning of NEE into its two flux components.

2.2.4. SiBCASA (Simple Biosphere/Carnegie-Ames-Stanford Approach)

SiBCASA (Schaefer et al., 2008; Van der Velde et al., 2014) combines parameterization of the Simple Biosphere model (SiB) with the biogeochemistry of the Carnegie-Ames Stanford Approach (CASA) and additional features. It can estimate terrestrial carbon fluxes from diurnal to decadal timescales by calculating water, carbon, and energy exchanges between 25 soil layers, plants, and the atmosphere. It derives photosynthesis rate using the Ball-Berry-Woodrow model of stomatal conductance (Ball et al., 1987), and C3 and C4 vegetation types are treated separately in the kinetic enzyme model of Farquhar et al. (1980). The simulation used here is forced using meteorological inputs from ERA-Interim, and run with 10-min time steps, 1 × 1 degrees spatial resolution, and hourly temporal resolution. SiBCASA uses observed LAI (Leaf Area Index) measurements to guide semi-prognostic canopy development; and in this study, we used the particular SiBCASA simulation focused on the 2018 drought.

2.3. Atmospheric Transport Simulations-LUMIA

 CO_2 is a stable tracer in the atmosphere: its sources and sinks are almost exclusively at the surface (small quantities are produced directly in the atmosphere by CO oxidation and emitted by aircraft, but these are negligible in comparison to CO_2 exchanges at the surface). Thus, the atmospheric distribution of CO_2 largely reflects the strengths and spatio-temporal distributions of surface CO_2 fluxes and subsequent mixing (convective and advective) of air. Thus, provided that the atmospheric mixing can be properly simulated, it is possible to relate surface CO_2 fluxes to observed CO_2 concentrations.

We implemented a regional transport model, following the approach used in the LUMIA system (Monteil & Scholze, 2021). The modeled mixing ratios y_m are defined as the sum of two contributions: a "foreground" (near-field) contribution, which is the impact of fluxes on CO₂ concentrations in the direct vicinity, spatial and temporal, of the observations, and a "background" (far-field) contribution, which is the contribution of fluxes further away in time and/or space. In LUMIA, the background contribution is computed using the global, coarse resolution (6° × 4°) transport model TM5, and the foreground using observation footprints of the FLEXPART Lagrangian particle dispersion model. For a given observation o, the modeled concentration y_m^o is given by

$$y_m^o = y_{bg}^o + \sum_c \sum_{i,j,t} K_{i,j,t}^o f_{i,j,t}^c$$

where y_{bg}^{o} is the estimated contribution of the background to observation o; $f_{i,j,t}^{c}$ is the CO₂ flux estimate for the flux category c ("anthropogenic emissions" or "NEE"), at the lat, long. and time coordinates (i, j, t), respectively, and K^{o} is the "footprint" storing the sensitivity of observation o to the surface fluxes $(K_{i,j,t}^{o} = \frac{dy^{o}}{df_{i,j,t}})$. The footprints K are in s.m² .mol(air)⁻¹, and the regional fluxes f in μ mol(CO₂).m⁻².s⁻¹, so their product is in

 μ mol(CO₂)/mol(air) (i.e., ppm CO₂, like the observations).



The observation footprints were computed using the FLEXPART 10.0 model, which simulates the dispersion backward in time of numerous virtual air particles (10,000 in our setup), based on meteorological input data. The aggregated residence time of the particles in a layer below 100 m above ground is taken as a proxy for the observations' sensitivity to the surface CO_2 exchanges. The footprints were computed at half degree spatial and three-hourly temporal resolution, and the FLEXPART model was driven by ECMWF ERA-Interim reanalysis data (Pisso et al., 2019). The footprints cover a regional domain ranging from 0°E, 54°N to 30°E, 70°N (Figure 1), and each covers a period up to 14 days before the observation time (although, in most cases, the sensitivity drops to zero in just a few days, as the particles all exit the regional domain).

The background concentrations (y_{bg}) were computed using the global, coarse resolution TM5-4DVAR inverse model, following the approach described in further detail by Monteil and Scholze (2021). Briefly, TM5 not only provides a boundary condition at the edge of the domain, but also computes the transport of that boundary condition toward the observation sites. There is therefore no exchange of masses between the two transport models; each calculates a partial mixing ratio. The TM5 simulation was driven at global, $6^{\circ} \times 4^{\circ}$, 3-hourly resolution, by ECMWF ERA-Interim meteorological data and transporting CO₂ fluxes. The latter were obtained from global TM5-4DVAR inversion assimilating flask observations from the NOAA ESRL Carbon Cycle Cooperative Air Sampling Network (Dlugokencky et al., 2013) and, within Europe, from the ICOS network of tall towers, in efforts to ensure that the background concentrations were as unbiased as possible. See Monteil and Scholze (2021) for more complete details.

3. Results

We compared the carbon flux estimates of the Northern Sweden region using vegetation models at regional scale and empirical observations of fluxes and concentration at site scales. Section 3.1 summarizes results of the comparison of fluxes from vegetation models with tall tower EC measurements to illustrate how well the vegetation models reproduced the observed ecosystem fluxes at the study site. In Section 3.2, we compare the atmospheric CO_2 concentrations simulated by LUMIA using the terrestrial fluxes simulated by the vegetation models with the observed atmospheric concentrations to assess the applicability of site-scale conclusions to regional scale patterns and thus facilitate evaluation of ecosystem models' performance.

3.1. Analysis of the In Situ Data

Seasonal variations in weekly sums of NEE, GPP, and Reco simulated by the four vegetation models are compared to tall tower EC measurements at Svartberget during the 2016–2018 study period in Figure 2. Flux estimates obtained using the LPJG Conif and LPJG PNV (see Section 2.2.1) are shown in this comparison, but other LPJ-GUESS modeling results presented in this paper are based on gridded simulations. Results of the two LPJ-GUESS site simulations and gridded LPJ-GUESS simulations at site and region levels are presented separately in Figure S2 in Supporting Information S1. Briefly, the fluxes obtained from the site simulations are similar to the one obtained from the gridded simulation, but have larger variability on daily timescales. We suspect this is because the landscape around the Svartberget site is fairly homogeneous. However, the locally measured meteorological parameters are more variable than the corresponding parameters in the gridded CRU-JRA meteo-data. Interestingly, the net fluxes from the gridded simulation are closer to the observed fluxes, possibly because the site simulations were not calibrated to the specific site conditions as explained above. According to both in the EC data and models, CO₂ was emitted (net release of CO₂ occurred) from October until the end of the following April, regarded here as winter period. This offset uptake during the spring and summer, from March until the following September, with a maximum rate of 50–100 gC m⁻² week⁻¹. Simulations by all models for all years showed the same general pattern, but emissions and uptake were stronger according to LPJ-GUESS and ORCHIDEE than the other models (Figure 2b). Modeled fluxes from VPRM appeared to be underestimated with lower CO, release in winter. Both VPRM and SiBCASA showed lower net CO, uptake in the summer of 2016, and it persisted for most of 2017 and 2018 for SiBCASA compared to the tall tower EC measurements. The differences between modeled and observed NEE were larger during winter periods (weekly model-data mismatch 20 gC m⁻²), than during summer (weekly model-data mismatch 10 gC m⁻²), and largest in the VPRM and LPJ-GUESS simulations. While VPRM simulated zero fluxes throughout the winter, fluxes were only non-zero in LPJ-GUESS simulations in the beginning of each year (January to March). All site-scale LPJ-GUESS simulations



Journal of Geophysical Research: Atmospheres



Figure 3. Comparison of annual cumulative net ecosystem carbon exchange (NEE), gross primary production (GPP), and ecosystem respiration (Reco) estimates by the four vegetation models (LPJ-GUESS, VPRM, SiBCASA, and ORCHIDEE; columns 1–4, left to right) and landscape level eddy covariance (EC data) measurements (far right column) at the Svartberget (SVB) site during 2016–2018.

show an emission peak in August 2018 with a mean model-data mismatch of 30 gC m⁻² with respect to the EC data. This summer emission peak in LPJ-GUESS simulations, which conflicts with simulations by the other models and the EC-based NEE observations, is largely due to substantially higher Reco values. In contrast, the EC data show a sharp increase in net CO₂ uptake, compared to the seasonal mean, in June 2018, due to an increase in GPP and decrease in Reco during this period (Figures 2b and 2c).

For the GPP and Reco component fluxes, differences between the simulations and EC data were more pronounced during the summer periods in the three study years. Like the LPJ-GUESS site-scale simulations, OR-CHIDEE overestimated GPP and Reco substantially during the summers. Essentially, the two process-based models (LPJ-GUESS and ORCHIDEE) overestimated both component fluxes with respect to the EC fluxes. Nevertheless simulated and recorded NEE fluxes are more similar. In stark contrast, VPRM underestimated the two component fluxes, but the simulated NEE matches the observed NEE well, except for zero winter fluxes. In addition, ORCHIDEE and SiBCASA simulated winter respiration and photosynthesis well, but not the other vegetation models. However, all vegetation models reproduced the observed fluxes and their seasonal dynamics within the model limitations during the entire annual period.

To explore the disparities in intra and interannual results of the applied models further, it is important to examine the net CO_2 exchange at landscape scale, the total annual sums and differences in spring onset as well as periods when gross CO_2 uptake via photosynthesis exceeds respiratory CO_2 release (or vice versa). These features can be explored with the help of cumulated curves of ecosystem fluxes (Figure 3) generated by the four vegetation models and tall tower EC measurements at SVB for the years 2016, 2017, and 2018. Both models and observations showed the same general seasonal patterns, which switch between winter net emission to summer net uptake in



May and October, respectively. However, there are some notable differences among the simulations generated by the four models and the EC data set. The interannual difference in NEE curves generated by all four models started around May. This marks the onset of the spring period and corresponds to the start of increased CO_2 uptake. According to the EC data, this onset is different between years, occurring in March in 2016 but later in subsequent years (Figure 3e). The simulations do not display this difference, so the models apparently have limited ability to capture the interannual variation in the onset of the vegetation period manifested in the EC data. In addition, the cumulated NEE curve based on EC data showed a sharp shift upward in September 2017 (Figure 3e) due to an increase in Reco, which is captured by the VPRM and SiBCASA models, but not the others (Figures 3b and 3c). Meanwhile, the annual sums obtained from the EC data and simulations agreed well considering the representation error of the vegetation models even though their validity cannot be compared, because of the use of data pertaining to a grid point in a regional simulation in the analysis. The summer and annual net CO_2 uptake were both lowest in 2017 (Figures 3b–3e) rather than during the drought year 2018 according to the EC data and most of the models, and highest in 2016, with VPRM yielding higher annual sums for all years than the other models. However, the LPJ-GUESS simulation suggested that annual net CO_2 uptake was lowest in 2018.

The seasonal and annual patterns of NEE, GPP, and Reco are more precisely captured (in terms of similarity to the EC data) by SiBCASA and ORCHIDEE (Figures 2 and 3). The magnitudes of daily NEE fluxes and annual sums simulated by LPJ-GUESS and VPRM are considerably higher and lower, respectively. The ecosystem model LPJ-GUESS and diagnostic model VPRM did not simulate the respiratory loss during winter indicated by the EC data. However, LPJ-GUESS generally reproduced the summer NEE uptake shown in the EC data well, except for the 2018 summer. The VPRM simulation did not capture the distinctive seasonal winter emission to summer uptake transition in any of the three years referring to the observations (Figure 3b). As shown in Figure 2, the component fluxes were respectively over- and underestimated by LPJ-GUESS and VPRM, but the balance between the two flux components yielded by LPJ-GUESS still resulted in comparable NEE fluxes to the EC estimates. The annual simulated NEE of 2017 differed from the EC data, but the modeled component fluxes generally agreed well, suggesting that GPP was higher and Reco lower in 2017 than in the other years. It is mentioned relative to the year-to-year difference, not referring to the absolute flux magnitude. In 2016, the GPP did not significantly differ from the GPP in other years and Reco was lower (which may account for the early transition to uptake in 2016). The daily NEE simulated by all models correlated well with the corresponding EC observations ($r^2 \ge 0.73$, RMSE ≤ 1.22 gC m⁻² day⁻¹) with the lowest linear Pearson correlation and highest RMSE for LPJ-GUESS ($r^2 = 0.73$, RMSE = 1.22 gC m⁻² day⁻¹) and highest correlation with lowest RMSE for VPRM and SiBCASA ($r^2 = 0.84$, RMSE = 1.09, and 1.04 gC m⁻² day⁻¹) (Table S1 in Supporting Information S1). Overall, the SiBCASA simulation provided the highest correlation with daily values of the observed EC fluxes (NEE, GPP and Reco) and lowest RMSE ($r^2 > 0.84$ and RMSE < 1.14 gC m⁻² day⁻¹; Tables S1 and S2 in Supporting Information S1) over the three study years. The annual GPP sums generated by the four models differ (Figure 3), but the linear correlation coefficients between modeled and EC estimates of daily GPP values are >0.88 with RMSE, which is <2.47 gC m⁻² day⁻¹. The modeled Reco values are less consistent with the observations (correlation coefficients ranging from 0.72 to 0.79 and RMSE 0.92-3.01 gC m⁻² day⁻¹). In addition, annual GPP was lower in 2017 and annual Reco higher in 2018 than in the other years (Figures 3 and 30) according to both the EC data and all models (Figures 3f-3i and 3k-3n).

3.2. Regional-Scale Analysis

In this section, we assess the potential utility of atmospheric CO_2 observations for validating bottom-up estimates of CO_2 fluxes at regional scale, and whether the site-scale features observed at SVB are representative of CO2 fluxes in Northern Sweden. We focus specifically on the "Northern Sweden" region (within the blue solid line in Figure 6), covering 320,000 km² around SVB. This region is regarded as fairly typical for much of the rest of Northern Sweden, and the measured concentrations at SVB are most sensitive to fluxes in this region. In Section 3.2.1, we considered the fit of results generated by the atmospheric transport model described in Section 2.3, driven by modeled CO_2 flux estimates, to CO_2 concentrations measured at ICOS sites in the Nordic study domain. It should be noted that the background concentration is always the major component of the total measured concentration at any location in the domain. The input ecosystem-driven fluxes only contribute significantly to changes in the relatively minor foreground concentration.





Figure 4. (a) Comparison of weekly averages of the hourly daytime LUMIA-simulated CO_2 concentration obtained using indicated vegetation models with observed concentrations (Obs) at the ICOS Svartberget (SVB) site. (b) Same as (a), but only for periods when the sensitivity of observations (of hourly daytime concentration) to fluxes of the Northern Sweden region exceeded 70%. The background (bg) concentration is shown in black.

3.2.1. Atmospheric CO, Concentration

The magnitude of weekly variation in daytime (11:00–14:00 UTC) CO_2 concentrations at SVB (Figure 4) was similar according to both the simulations and observations. We focused on daytime values because the transport model is expected to be most accurate when the planetary boundary layer (PBL) is well developed. We considered weekly means partly for consistency across the flux and concentration analysis and partly because the weekly averages exhibit the seasonal pattern more clearly with less noise compared to the daily averages. Hence, they give a better appreciation of the overall model performance whereas shorter-time averages are relatively more sensitive to transport model errors and of the fine-scale performance of the vegetation models, especially for the 3-year study period (as illustrated by daily concentration in Figure S3 in Supporting Information S1). The Pearson correlation coefficients between observed and simulated CO_2 concentrations always exceeded 0.93 with RMSE below 3.42 ppm for daytime hours at SVB (Table S3 in Supporting Information S1). However, the correlation coefficients were ca. 0.2 lower for the winters than for the summers. The high Pearson correlations between the simulations confirm the quality of the transport model and background concentration data, especially during daytime and summer.

The observed and modeled concentrations were both lower than the background concentration during summer, owing to CO_2 uptake, and vice versa during winter due to CO_2 release. On average, over the entire study period (2016–2018), VPRM-based estimates of CO_2 concentrations were closest to the observations, with less than ± 5 ppm bias and higher than 0.92 Pearson correlation for all sites in the study domain. SiBCASA and ORCHIDEE occasionally over-/underestimated concentrations, relative to observations, by more than ± 5 ppm sometimes. The unusual emission spike in the LPJ-GUESS simulation during August 2018 (Figure 2) at the SVB site was also manifested in the LPJ-GUESS-driven LUMIA simulation (2–4 ppm higher concentration than the observations during 2018 summer) while other LUMIA simulations closely followed the observed concentration throughout the study period. Apart from this, all models provided better fits to the observations during 2018 than the previous year, 2017. The near-zero wintertime fluxes in LPJ-GUESS and VPRM simulations led to underestimation of winter CO_2 concentration during the period 2017-09 to 2018-04 (Figure 4). As the background concentration is





Figure 5. Comparison of the weekly model to observation mismatch (modeled-observed CO_2 concentration) at indicated ICOS atmosphere sites in the study domain. It should be noted that data availability varies among sites.

the major component of the total measured concentration, it accounts for some of the concentration peaks, such as the one in November 2018. Weekly mean model-data mismatches (modeled minus observed concentrations) at all ICOS atmospheric sites in the study domain are shown in Figure 5. The weekly biases at SVB in simulations by all the models stayed within -10 to 6 ppm (Figure 5a), while at other sites, it varied between -10 and 10 ppm. On average across all sites, there was a weekly bias of 3 ppm in LPJ-GUESS and VPRM simulations, but in some cases (e.g., the summer of 2018 at SVB, PAL, and PUI) LPJ-GUESS estimates were up to 5 ppm higher than the observed concentrations. Estimates by the other models had a negative bias of up to -8 ppm at SVB. ORCHIDEE and SiBCASA simulations had a more positive bias than the others throughout the study period and overestimated the weekly concentrations by up to 10 ppm in winter. It should also be noted that concentration estimates across all the sites had higher bias in winter. This comparison of the transport models' correlations with atmospheric observations suggests that flux estimates by the diagnostic models, especially VPRM, are the most realistic at the regional scale (The correlations are shown in Table S3 in Supporting Information S1).

3.2.2. Regional CO₂ Fluxes

We investigated the strength and spatial distribution of annual NEE over the study domain to explore the representativeness of flux estimates at SVB for fluxes in the entire Northern Sweden region. Figure 6 illustrates the annual NEE estimated by the four vegetation models for the drought year 2018 over the Nordic domain. As already noted, VPRM indicated that both the site and the entire study region were larger sinks in 2018 than the other models. All of the models except ORCHIDEE agreed that the southern part of the domain was a larger sink in 2018 than the northern part. Although there were differences in the annual NEE simulated by the models, the distribution of NEE in each simulation within the Northern Sweden region was homogenous.

The relative differences in annual NEE at the site (landscape) and regional (Northern Sweden) scales for each of the three study years are shown in Figure 7. All models except SiBCASA suggested that both the site and region were a carbon sink during the three years, although the estimated sink strength varied from -275 to -25 gC m⁻² year⁻¹. The site and region were smaller sinks according to all the models and EC data (or even a small source according to the SiBCASA simulation) in 2017 than in 2018 and 2016, except the LPJ-GUESS simulation, in which the sink was smallest in 2018. NEE budgets obtained from ORCHIDEE and VPRM were quite constant from year to year with standard deviation ± 6.6 and ± 19.3 gC m⁻² year⁻¹, respectively. The substantial discrepancy in annual NEE for 2018 at site scale between the LPJ-GUESS and other simulations resulted from the summer emission peak it generated at SVB site (Figure 2). However, this site-scale feature was not prominent at the regional scale (see also Figure S2 in Supporting Information S1), and the annual NEE in LPJ-GUESS simulations





Figure 6. Spatial map of the annual net ecosystem exchange budget for 2018 derived using the four vegetation models (VPRM, LPJ-GUESS, ORCHIDEE, and SiBCASA). The Svartberget (SVB) site is marked with a blue star and the Northern Sweden region is outlined in blue.

is very similar to that of ORCHIDEE and SiBCASA simulations for 2018 at the regional scale. Except for this peak in the LPJ-GUESS simulations, the NEE dynamics and annual budgets at the SVB site are similar to those for the rest of the Northern Sweden domain in the simulations by all models and the model-to-model differences are larger than the year-to-year differences within each simulation.

We included all available hourly daytime observations when calculating the weekly averages at SVB presented in Figure 4a. However, CO_2 observations at SVB are not equally sensitive to fluxes from within the Northern Sweden region. Depending on the wind intensity and direction, they can be more sensitive to other parts of the domain or the background concentration. Thus, we used the FLEXPART footprints (see Section 2.3) to filter



Figure 7. Bar plots showing annual sums of Net Ecosystem Exchange (NEE in of gC m^{-2} year⁻¹) for three years (2016–2018) at the SVB site and the Northern Sweden region from four models and tall tower eddy covariance (EC) data.





Figure 8. Fractional sensitivity of indicated sites across Scandinavia to CO₂ fluxes in the Northern Sweden region (weekly means).

out the observations that were least sensitive to fluxes in the Northern Sweden region. For each observation, we computed a "regional sensitivity index":

$$FP_{cum} = \frac{\sum_{i_{reg}} K_{i_{reg}}}{\sum_{i} K_{i}} \times 100$$

where *K* is the FLEXPART footprint capturing sensitivities of the observation to each flux grid point (in time and space). The i_{reg} indices refer to the space/time grid cells within the region of interest (Northern Sweden), while the *i* indices refer to the ensemble of land grid-cells. The larger the FP_{cum}, the more sensitive the observation to fluxes in Northern Sweden (regardless of the region's eventual contribution to the modeled concentration, which depends on actual values of the fluxes).

We arbitrarily used a threshold value of $FP_{cum} \le 70\%$ rejecting observations, set as a compromise between selecting observations with close to maxima local influence and not rejecting too many observations. The interpretation is therefore mostly qualitative. We show weekly averages of the modeled and observed concentrations in Figure 4b, for the observations that met this criterion. The influence of regional fluxes was most prominent during some peaks, particularly in August 2017 and March 2018.

To determine how a network of observation sites could help us to further constrain the fluxes in Northern Sweden, we repeated the same "regional sensitivity index" calculation for the other ICOS sites in our domain. Figure 8 shows time courses of that fractional sensitivity metric for all the sites. Apart from SVB, the sites that were most sensitive to fluxes in the region are Pallas (PAL) and Norunda (NOR), and they were particularly sensitive during the times of the unusually high emissions in the LPJ-GUESS simulation at SVB in August 2018 and August 2017. However, these observations were also very sensitive to fluxes from their own region and other regions, which are uncertain. This complicates use of these observations for direct validation of the fluxes, but indicates that they could potentially provide important constraints in a formal inverse modeling approach.

4. Discussion

The carbon balance of boreal regions is of particular interest due to their high sensitivity to warming and high rates of projected warming (Ito et al., 2020; Vincent, 2020). Thus, we have explored in detail the agreement between CO_2 fluxes (NEE and its component fluxes GPP and Reco) obtained from vegetation models and observations, including their seasonal patterns and interannual variability, over Northern Sweden. We compared gross and net fluxes obtained using four well-established vegetation models (LPJ-GUESS, ORCHIDEE, VPRM and SiBCASA) to constrain the ecosystem carbon budgets at both the site and regional scale. All models unequivocally indicate that the region was a carbon sink during the 3-year study period (2016–2018). Here, we focus on



three aspects of the study: the difference in estimates among models and observation, the potential utility of ICOS observations for constraining fluxes in our domain, and the accuracy of the models' predictions for fluxes during the drought year 2018.

4.1. Disparities Among Estimates of Carbon Exchange in Northern Sweden Obtained From the Vegetation Models and Observations

There are substantial differences in the annual budgets estimated by the four models, largely due to differences in their internal structures and dynamics. The relative contributions of uptake and emission in the simulations differ from the observations at landscape scale, resulting in differences of 100-200 gC m⁻² year⁻¹ in NEE annual budgets. The VPRM and LPJ-GUESS models simulated the highest and second highest annual ecosystem carbon budgets generally. High carbon uptake is a known feature of VPRM, resulting from the model parameters' calibration against EC observations (Oney et al., 2017). The near-zero winter flux simulated by VPRM is not consistent with either the other models or tall tower EC fluxes. Since photosynthetic rates are extremely low in winter (so GPP is very close to zero), ecosystem respiration (Reco) continues and NEE is positive. The winter fluxes are more realistically represented by SiBCASA and ORCHIDEE, and the fluxes simulated by these models are closest to the empirically determined EC fluxes. This holds true for the component fluxes also from SiBCA-SA which compares best with EC fluxes for the whole study period despite estimating the lowest annual NEE budget among other models. In high-latitude forests, carbon uptake may generally be promoted more strongly by the higher radiation levels associated with drought than it is impaired by the decline in water availability (Flach et al., 2018), unless the drought is extremely severe. Our findings suggest that the considered model simulations of NEE fluxes differ significantly from those manifested in the EC data, although both the interannual and seasonal variations were quite similar. On the other hand, the EC flux components, GPP and Reco, are also estimates generated by flux partitioning algorithms from direct measures of NEE obtained using the EC technique. Thus, use of a source partitioning model in the prediction of fluxes may also contribute to uncertainties. We found that the sharp reduction in net CO₂ uptake in September 2017 recorded only in EC observations was caused by a rainy period (with 80 mm precipitation between 7 and 13th September 2017), in which global radiation (and hence GPP) was considerably reduced, while the rewetting likely increased Reco (see, Figure 30). Since we only used the gridded simulations here with meteorological inputs from a regional-scale product rather than local observations of meteorological variables at SVB, it is not surprising that the models did not simulate any responses to such short-term weather events, which got smoothed out.

4.2. Can We Use an Atmospheric Observation Network to Constrain Fluxes in Northern Sweden?

As mentioned, one of the objectives of the study was to assess ICOS observations' potential utility for constraining surface CO_2 fluxes in Northern Europe, the study domain. The density of the observation network is limited, but the region has relatively homogeneous ecosystems (with north-south climate gradients), and fewer major urban areas than other parts of Europe. Thus, it is a good candidate area for estimating carbon fluxes at large scales using an inverse approach. The relative homogeneity of the flux patterns simulated by the four vegetation models confirms that the density of the network is not a fundamental obstacle (unless there is more variability in the fluxes than the terrestrial biosphere models indicate). Another potential obstacle is that the contribution of regional fluxes to the observed concentrations could be within the uncertainty of the background concentrations. However, the CO_2 transport simulations presented in Section 3.2 show that the atmospheric observations have good sensitivity to regional signals. The differences between the flux estimates lead to differences in concentrations of the order of 10–20 ppm (Figure 5), which is significantly larger than the typical uncertainty of the observations and background concentrations (<1 ppm). The estimated regional sensitivities in Section 3.2.2 (Figure 8) corroborate our conclusion that the observation network provides exploitable constraints, and we will implement a formal inversion approach in further research.

4.3. Performance of Vegetation Models During the 2018 Summer Drought

A negative effect of the 2018 drought on the NEE of Northern Sweden was observed, but it was less pronounced than its effect in the south of Sweden and other parts of Europe (Gharun et al., 2020; Rödenbeck et al., 2020; Peters et al., 2020; Lindroth et al., 2020). However, the NEE response to drought simulated by the models differed,



with LPJ-GUESS simulation showing a net reduction in NEE uptake for 2018 at the SVB site, but not the observations or other models. The EC data showed an increase in net uptake in the early summer of 2018, which led to an annual budget similar to that of the "normal" reference year 2016, in which precipitation was near-normal across most of Europe (Fu et al., 2020). The observed annual carbon sink was lower in 2017, because in that year a late cold spring was followed by a cold and cloudy summer, resulting in lower than normal CO₂ uptake (Chi et al., 2019). Apart from the increased uptake shown in EC data and high emissions in LPJ-GUESS simulations in the summer of 2018, the general trend in the simulations indicated that annual net CO_2 uptake was lower in 2018 than the reference year except SiBCASA. The lower annual uptake reflects the limited impact of the dry summer, which was partly compensated by the colder start of the year and warmer spring (Gharun et al., 2020) than in 2016 at site and regional scale. Although 2018 was one of the warmest years on record (Bastos et al., 2020; Thompson et al., 2020; Figure S1 in Supporting Information S1), the drought did not have a strong impact on the annual carbon balance of Northern Sweden. Buras et al. (2019) obtained similar results and attributed the lesser impact of the 2018 drought to weaker sensitivity of the boreal mixed land-cover types in Northern Sweden. In addition, adverse effects of the 2018 drought were mediated more strongly by low soil water content (SWC) than high air temperature (Fu et al., 2020). Hence, it was likely to have relatively weak effects in the northern Nordic forests where the trees generally have relatively high rooting depths and access to deeper soil water resources than those in many other ecosystems (Teuling et al., 2010).

The spatial plots clearly show a sharp north-south gradient in ecosystem responses to the drought, as also reported in Bastos et al. (2020). The simulated NEE responses generally agreed well with previous reports based on NEE measurements (Lindroth et al., 2020), not only for Northern Sweden, but also for the whole of Northern Fennoscandia. However, the apparent discrepancies call for further efforts to improve model algorithms to enhance the accuracy of predicted carbon-cycle climate feedbacks under future climate change.

5. Conclusions

Knowledge of Nordic ecosystems' responses to climate change is crucial for assessing their potential contributions to mitigation efforts. To facilitate acquisition of such knowledge, we have established a framework to combine observations of fluxes and concentrations with data generated by vegetation and atmospheric transport models which enhance estimates of the carbon balance of Northern Sweden. We also assessed the SVB experimental site's representativeness for the region. We found that NEE estimates from four vegetation models and tall tower EC measurements generally agree that the region was a carbon sink during the three-year study period (2016– 2018). However, simulations by the four tested models differed in estimates of the sink strength and responses to the 2018 drought. They also did not perform well in reproducing the inter- and intra-annual variability and seasonal dynamics. Based on consistency of the simulations and observations of flux and CO₂ concentrations, we conclude that the 2018 drought had limited impact on the Northern Sweden region's capacity to sequester carbon, although it had severe effects in southern Sweden, Denmark, and Estonia (Lindroth et al., 2020). The high correlations between modeled and observed daytime concentrations demonstrate the high quality of the transport model and background concentration estimates used in the study, especially during daytime and summer. In addition, our regional sensitivity analysis confirmed that the source area/region of fluxes affect the measured concentrations. Thus, our study indicates that the ICOS atmospheric observation network has substantial potential utility for validating the modeled fluxes in the region in an inversion framework. However, we also noted discrepancies that may at least partly reflect a need for further refinement of these general vegetation models for use in Nordic countries, and their evaluation at other ICOS sites to improve their flux estimates for transport models. Overall, we conclude that the integration of vegetation and transport models with tall tower flux and concentration measurements is a promising approach for reconciling regional carbon balances and their responses to climate change.

Data Availability Statement

The modeled concentrations, cumulated sensitivities, background concentrations, etc as well as the regions definition are shared via the figshare link (https://figshare.com/articles/dataset/Data/16676656).



Acknowledgments

The first author, Anusha Sathyanadh, acknowledges a Postdoctoral Scholarship funded by the Kempe Foundations (grant no: JCK-1815). A grant (no. 942-2015-49) from the Swedish Research Council for Environment, Agricultural Sciences and Spatial Planning (FORMAS) funded the tall tower eddy covariance measurements at Svartberget. Financial support from the Swedish Research Council and research institutes contributing to the Swedish Integrated Carbon Observation System (ICOS-Sweden) and Research Infrastructure from the Swedish Infrastructure for Ecosystem Science (SITES) is also acknowledged. Marko Scholze and Guillaume Monteil acknowledge support from the three Swedish strategic research areas ModElling the Regional and Global earth system (MERGE), the e-science collaboration (eSSENCE), and Biodiversity and Ecosystems in a Changing Climate (BECC). The computations were enabled with resources provided by the Swedish National Infrastructure for Computing (SNIC) at NSC partially funded by the Swedish Research Council through grant agreement no. 2018-05973. We thank Greet Janssens-Meanhout (European Commission, Joint Research Center, 790 Ispra, Italy) for providing the fuel type and category specific version of the EDGAR v4.3 anthropogenic emission data and Hugo Denier van der Gon (TNO, The Netherlands) for making the temporal emission profiles available. We also thank the PIs of the atmospheric stations at ICOS Svartberget, Norunda, Hyltemossa, Puijo, Pallas, and Hyytiala (SE-SVB, SE-NOR, SE-HTM, FI-PUI, FI-PAL, and FI-SMR) for providing data on atmospheric CO₂ concentrations. The authors thank the staff at the SLU Unit for Field-Based Forest Research for technical and logistic support at the Svartberget station.

References

- Alekseychik, P., Mammarella, I., Karpov, D., Dengel, S., Terentieva, I., Sabrekov, A., et al. (2017). Net ecosystem exchange and energy fluxes measured with the eddy covariance technique in a western Siberian bog. *Atmospheric Chemistry and Physics*, 17(15), 9333–9345. https://doi. org/10.5194/acp-17-9333-2017
- Amthor, J. S., Chen, J. M., Clein, J. S., Frolking, S. E., Goulden, M. L., Grant, R. F., et al. (2001). Boreal forest CO2 exchange and evapotranspiration predicted by nine ecosystem process models: Intermodel comparisons and relationships to field measurements. *Journal of Geophysical Research*, 106(D24). https://doi.org/10.1029/2000JD900850
- Baldocchi, D., Falge, E., & Wilson, K. (2001). A spectral analysis of biosphere–atmosphere trace gas flux densities and meteorological variables across hour to multi-year time scales. Agricultural and Forest Meteorology, 107(1), 1–27. https://doi.org/10.1016/s0168-1923(00)00228-8
- Baldocchi, D. D. (2003). Assessing the eddy covariance technique for evaluating carbon dioxide exchange rates of ecosystems: Past, present and future. *Global Change Biology*, 9(4), 479–492. https://doi.org/10.1046/j.1365-2486.2003.00629.x
- Ball, J. T., Woodrow, I. E., & Berry, J. A. (1987). A model predicting stomatal conductance and its contribution to the control of photosynthesis under different environmental conditions. In *Progress in photosynthesis research* (pp. 221–224). Springer. https://doi. org/10.1007/978-94-017-0519-6_48
- Ballantyne, A. P., Liu, Z., Anderegg, W. R. L., Yu, Z., Stoy, P., Poulter, B., et al. (2021). Reconciling carbon-cycle processes from ecosystem to global scales. *Frontiers in Ecology and the Environment*, 19(1), 57–65. https://doi.org/10.1002/fee.2296
- Barber, V. A., Juday, G. P., & Finney, B. P. (2000). Reduced growth of Alaskan white spruce in the twentieth century from temperature-induced drought stress. *Nature*, 405(6787), 668–673. https://doi.org/10.1038/35015049
- Bastos, A., Fu, Z., Ciais, P., Friedlingstein, P., Sitch, S., Pongratz, J., et al. (2020). Impacts of extreme summers on European ecosystems: A comparative analysis of 2003, 2010 and 2018. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1810), 20190507. https://doi.org/10.1098/rstb.2019.0507
- Bradshaw, C. J. A., & Warkentin, I. G. (2015). Global estimates of boreal forest carbon stocks and flux. *Global and Planetary Change*, 128, 24–30. https://doi.org/10.1016/j.gloplacha.2015.02.004
- Brandt, J. P., Flannigan, M. D., Maynard, D. G., Thompson, I. D., & Volney, W. J. A. (2013). An introduction to Canada's boreal zone: Ecosystem processes, health, sustainability, and environmental issues. *Environmental Reviews*, 21(4), 207–226. https://doi.org/10.1139/er-2013-0040
- Braswell, B. H., Schimel, D. S., Linder, E., & Moore, B. (1997). The response of global terrestrial ecosystems to interannual temperature variability. Science, 278(5339), 870–873. https://doi.org/10.1126/science.278.5339.870
- Buras, A., Rammig, A., & Zang, C. S. (2019). Quantifying impacts of the drought 2018 on European ecosystems in comparison to 2003. *Biogeosciences Discussions* (August), 1–23. https://doi.org/10.5194/bg-2019-286
- Campioli, M., Malhi, Y., Vicca, S., Luyssaert, S., Papale, D., Peñuelas, J., et al. (2016). Evaluating the convergence between eddy-covariance and biometric methods for assessing carbon budgets of forests. *Nature Communications*, 7(1), 1–12. https://doi.org/10.1038/ncomms13717
- Chapin, F. S., Woodwell, G. M., Randerson, J. T., Rastetter, E. B., Lovett, G. M., Baldocchi, D. D., et al. (2006). Reconciling carbon-cycle concepts, terminology, and methods. *Ecosystems*, 9(7), 1041–1050. https://doi.org/10.1007/s10021-005-0105-7
- Chi, J., Nilsson, M. B., Kljun, N., Wallerman, J., Fransson, J. E. S., Laudon, H., et al. (2019). The carbon balance of a managed boreal landscape measured from a tall tower in northern Sweden. Agricultural and Forest Meteorology, 274(March), 29–41. https://doi.org/10.1016/j. agrformet.2019.04.010
- Chi, J., Nilsson, M. B., Laudon, H., Lindroth, A., Wallerman, J., Fransson, J. E. S., et al. (2020). The net landscape carbon balance—Integrating terrestrial and aquatic carbon fluxes in a managed boreal forest landscape in Sweden. *Global Change Biology*, 26(4), 2353–2367. https://doi. org/10.1111/gcb.14983
- Ciais, P., Reichstein, M., Viovy, N., Granier, A., Ogée, J., Allard, V., et al. (2005). Europe-wide reduction in primary productivity caused by the heat and drought in 2003. *Nature*, 437(7058), 529–533. https://doi.org/10.1038/nature03972
- Cohen, J., Screen, J. A., Furtado, J. C., Barlow, M., Whittleston, D., Coumou, D., et al. (2014). Recent Arctic amplification and extreme mid-latitude weather. *Nature Geoscience*, 7(9), 627–637. https://doi.org/10.1038/ngeo2234
- Collatz, G. J., Ball, J. T., Grivet, C., & Berry, J. A. (1991). Physiological and environmental regulation of stomatal conductance, photosynthesis and transpiration: A model that includes a laminar boundary layer. Agricultural and Forest Meteorology, 54(2–4), 107–136.
- Cramer, W., Bondeau, A., Woodward, F. I., Prentice, I. C., Betts, R. A., Brovkin, V., et al. (2001). Global response of terrestrial ecosystem structure and function to CO₂ and climate change: Results from six dynamic global vegetation models. *Global Change Biology*, 7(4), 357–373. https://doi.org/10.1046/j.1365-2486.2001.00383.x
- Dlugokencky, E., Crotwell, A., Masarie, K., White, J., Lang, P., & Crotwell, M. (2013). NOAA measurements of long-lived greenhouse gases. Asia-Pacific GAW Greenhouse Gases, 6, 6–9.
- Farquhar, G. D., von Caemmerer, S., & Berry, J. A. (1980). A biochemical model of photosynthetic CO₂ assimilation in leaves of C3 species. *Planta*, 149(1), 78–90. https://doi.org/10.1007/bf00386231
- Fauré, E., Finnveden, G., & Gunnarsson-Östling, U. (2019). Four low-carbon futures for a Swedish society beyond GDP growth. Journal of Cleaner Production, 236, 117595.
- Flach, M., Sippel, S., Gans, F., Bastos, A., Brenning, A., Reichstein, M., et al. (2018). Contrasting biosphere responses to hydrometeorological extremes: Revisiting the 2010 western Russian heatwave. *Biogeosciences*, 15(20), 6067–6085. 10.5194/bg-15-6067-2018
- Friend, A. D., Arneth, A., Kiang, N. Y., Lomas, M., Ogée, J., Rödenbeck, C., et al. (2007). FLUXNET and modelling the global carbon cycle. *Global Change Biology*, 13(3), 610–633. https://doi.org/10.1111/j.1365-2486.2006.01223.x
- Fu, Z., Ciais, P., Bastos, A., Stoy, P. C., Yang, H., Green, J. K., et al. (2020). Sensitivity of gross primary productivity to climatic drivers during the summer drought of 2018 in Europe. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1810), 20190747. https:// doi.org/10.1098/rstb.2019.0747
- Gauthier, S., Bernier, P., Kuuluvainen, T., Shvidenko, A. Z., & Schepaschenko, D. G. (2015). Boreal forest health and global change. *Science*, 349(6250), 819–822. https://doi.org/10.1126/science.aaa9092
- Gharun, M., Hörtnagl, L., Paul-Limoges, E., Ghiasi, S., Feigenwinter, I., Burri, S., et al. (2020). Physiological response of Swiss ecosystems to 2018 drought across plant types and elevation. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1810), 20190521. https://doi.org/10.1098/rstb.2019.0521
- Gurney, K. R., Law, R. M., Denning, A. S., Rayner, P. J., Baker, D., Bousquet, P., et al. (2002). Towards robust regional estimates of CO 2 sources and sinks using atmospheric transport models. *Nature*, 415(6872), 626–630. https://doi.org/10.1038/415626a
- Haas, E., Klatt, S., Fröhlich, A., Kraft, P., Werner, C., Kiese, R., et al. (2013). LandscapeDNDC: A process model for simulation of biosphere-atmosphere-hydrosphere exchange processes at site and regional scale. *Landscape Ecology*, 28(4), 615–636. https://doi.org/10.1007/s10980-012-9772-x

- Hurtt, G. C., Chini, L. P., Frolking, S., Betts, R. A., Feddema, J., Fischer, G., et al. (2011). Harmonization of land-use scenarios for the period 1500–2100: 600 years of global gridded annual land-use transitions, wood harvest, and resulting secondary lands. *Climatic Change*, 109(1–2), 117. https://doi.org/10.1007/s10584-011-0153-2
- ICOS, R. I. (2019). ICOS atmospheric greenhouse gas mole fractions of CO₂, CH₄, CO, ¹⁴CO₂ and meteorological observations september 2015– April 2019 for 19 stations (49 vertical levels), final quality controlled level 2 data.
- Ito, A., Reyer, C. P. O., G\u00e4deke, A., Ciais, P., Chang, J., Chen, M., et al. (2020). Pronounced and unavoidable impacts of low-end global warming on northern high-latitude land ecosystems. *Environmental Research Letters*, 15(4), 44006. https://doi.org/10.1088/1748-9326/ab702b
- Janssens-Maenhout, G., Crippa, M., Guizzardi, D., Muntean, M., Schaaf, E., Dentener, F., et al. (2019). EDGAR v4. 3.2 Global Atlas of the three major greenhouse gas emissions for the period 1970–2012. *Earth System Science Data*, 11(3), 959–1002. https://doi.org/10.5194/ essd-11-959-2019
- Jung, M., Henkel, K., Herold, M., & Churkina, G. (2006). Exploiting synergies of global land cover products for carbon cycle modeling. *Remote Sensing of Environment*, 101(4), 534–553. https://doi.org/10.1016/j.rse.2006.01.020
- Jung, M., Le Maire, G., Zaehle, S., Luyssaert, S., Vetter, M., Churkina, G., et al. (2007). Assessing the ability of three land ecosystem models to simulate gross carbon uptake of forests from boreal to Mediterranean climate in Europe.
- Kasischke, E. S. (2000). Boreal ecosystems in the global carbon cycle. In *Fire, climate change, and carbon cycling in the boreal forest* (pp. 19–30). Springer. https://doi.org/10.1007/978-0-387-21629-4_2
- Kondo, M., Patra, P. K., Sitch, S., Friedlingstein, P., Poulter, B., Chevallier, F., et al. (2020). State of the science in reconciling top-down and bottom-up approaches for terrestrial CO2 budget. *Global Change Biology*, 26(3), 1068-1084. https://doi.org/10.1111/gcb.14917
- Kountouris, P., Gerbig, C., Totsche, K.-U., Dolman, A.-J., Meesters, A.-G.-C.-A., Broquet, G., et al. (2015). An objective prior error quantification for regional atmospheric inverse applications. *Biogeosciences*, 12(24), 7403–7421. https://doi.org/10.5194/bg-12-7403-2015
- Kramer, K., Leinonen, I., Bartelink, H. H., Berbigier, P., Borghetti, M., Bernhofer, C., et al. (2002). Evaluation of six process-based forest growth models using eddy-covariance measurements of CO2 and H2O fluxes at six forest sites in Europe. *Global Change Biology*, 8(3), 213–230. https://doi.org/10.1046/j.1365-2486.2002.00471.x
- Krinner, G., Viovy, N., de Noblet-Ducoudré, N., Ogée, J., Polcher, J., Friedlingstein, P., et al. (2005). A dynamic global vegetation model for studies of the coupled atmosphere-biosphere system. *Global Biogeochemical Cycles*, 19(1). https://doi.org/10.1029/2003gb002199
- Law, B. E., Falge, E., Gu, L. V, Baldocchi, D. D., Bakwin, P., Berbigier, P., et al. (2002). Environmental controls over carbon dioxide and water vapor exchange of terrestrial vegetation. Agricultural and Forest Meteorology, 113(1–4), 97–120. https://doi.org/10.1016/s0168-1923(02)00104-1
- Laudon, H., Maher Hasselquist, E., Peichl, M., Lindgren, K., Sponseller, R., Lidman, F., et al. (2021). Northern landscapes in transition: Evidence, approach and ways forward using the Krycklan Catchment Study. *Hydrological Processes*, 35(4), e14170. https://doi.org/10.1002/hyp.14170
- Law, B. E., Waring, R. H., Anthoni, P. M., & Aber, J. D. (2000). Measurements of gross and net ecosystem productivity and water vapour exchange of a Pinus ponderosa ecosystem, and an evaluation of two generalized models. *Global Change Biology*, 6(2), 155–168. https://doi. org/10.1046/j.1365-2486.2000.00291.x
- Lindroth, A., Holst, J., Linderson, M.-L., Aurela, M., Biermann, T., Heliasz, M., et al. (2020). Effects of drought and meteorological forcing on carbon and water fluxes in Nordic forests during the dry summer of 2018. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1810), 20190516. https://doi.org/10.1098/rstb.2019.0516
- Lindström, G., Bishop, K., & Löfvenius, M. O. (2002). Soil frost and runoff at Svartberget, northern Sweden Measurements and model analysis. *Hydrological Processes*, 16(17), 3379–3392. https://doi.org/10.1002/hyp.1106
- Mahadevan, P., Wofsy, S. C., Matross, D. M., Xiao, X., Dunn, A. L., Lin, J. C., et al. (2008). A satellite-based biosphere parameterization for net ecosystem CO2 exchange: Vegetation Photosynthesis and Respiration Model (VPRM). *Global Biogeochemical Cycles*, 22(2). https://doi. org/10.1029/2006GB002735
- Mahecha, M. D., Reichstein, M., Jung, M., Seneviratne, S. I., Zaehle, S., Beer, C., et al. (2010). Comparing observations and process-based simulations of biosphere-atmosphere exchanges on multiple timescales. *Journal of Geophysical Research: Biogeosciences*, 115(G2). https:// doi.org/10.1029/2009jg001016
- Malmsheimer, R. W., Bowyer, J. L., Fried, J. S., Gee, E., Izlar, R., Miner, R. A., et al. (2011). Managing forests because carbon matters: Integrating energy, products, and land management policy. *Journal of Forestry*, 109(7S), S7–S50.
- Mcguire, A. D., Anderson, L. G., Christensen, T. R., Scott, D., Laodong, G., Hayes, D. J., et al. (2009). Sensitivity of the carbon cycle in the Arctic to climate change. *Ecological Monographs*, 79(4), 523–555. https://doi.org/10.1890/08-2025.1
- Melillo, J. M., McGuire, A. D., Kicklighter, D. W., Moore, B., Vorosmarty, C. J., & Schloss, A. L. (1993). Global climate change and terrestrial net primary production. *Nature*, 363(6426), 234–240. https://doi.org/10.1038/363234a0
- Mittler, R., & Blumwald, E. (2010). Genetic engineering for modern agriculture: Challenges and perspectives. Annual Review of Plant Biology, 61, 443–462. https://doi.org/10.1146/annurev-arplant-042809-112116
- Monteil, G., Broquet, G., Scholze, M., Lang, M., Karstens, U., Gerbig, C., et al. (2020). The regional European atmospheric transport inversion comparison, EUROCOM: First results on European-wide terrestrial carbon fluxes for the period 2006–2015. Atmospheric Chemistry and Physics, 20(20), 12063–12091
- Monteil, G., & Scholze, M. (2021). Regional CO 2 inversions with LUMIA, the Lund University modular inversion algorithm, v1. 0. Geoscientific Model Development, 14(6), 3383–3406. https://doi.org/10.5194/gmd-14-3383-2021
- Moorcroft, P. R. (2006). How close are we to a predictive science of the biosphere? *Trends in Ecology & Evolution*, 21(7), 400–407. https://doi.org/10.1016/j.tree.2006.04.009
- Oney, B., Gruber, N., Henne, S., Leuenberger, M., & Brunner, D. (2017). A CO-based method to determine the regional biospheric signal in atmospheric CO2. *Tellus Series B Chemical and Physical Meteorology*, 69(1), 1–25. https://doi.org/10.1080/16000889.2017.1353388
- Pachauri, R. K., Allen, M. R., Barros, V. R., Broome, J., Cramer, W., Christ, R., et al. (2014). Climate change 2014: Synthesis report. Contribution of working groups I, II and III to the fifth assessment report of the intergovernmental panel on climate change. IPCC.
- Pan, S., Tian, H., Dangal, S. R. S., Ouyang, Z., Tao, B., Ren, W., et al. (2014). Modeling and monitoring terrestrial primary production in a changing global environment: Toward a multiscale synthesis of observation and simulation. Advances in Meteorology. https://doi. org/10.1155/2014/965936
- Pan, Y., Birdsey, R. A., Fang, J., Houghton, R., Kauppi, P. E., Kurz, W. A., et al. (2011). A large and persistent carbon sink in the world's forests. Science, 333(6045), 988–993. https://doi.org/10.1126/science.1201609
- Papale, D., Reichstein, M., Aubinet, M., Canfora, E., Bernhofer, C., Kutsch, W., et al. (2006). Towards a standardized processing of Net Ecosystem Exchange measured with eddy covariance technique: Algorithms and uncertainty estimation. *Biogeosciences*, 3(4), 571–583. https://doi. org/10.5194/bg-3-571-2006
- Peng, C. (2000). From static biogeographical model to dynamic global vegetation model: A global perspective on modelling vegetation dynamics. *Ecological Modelling*, 135(1), 33–54. https://doi.org/10.1016/s0304-3800(00)00348-3



- Peters, W., Bastos, A., Ciais, P., & Vermeulen, A. (2020). A historical, geographical and ecological perspective on the 2018 European summer drought. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375, 20190505. https://doi.org/10.1098/rstb.2019.0505
- Petrescu, A. M. R., McGrath, M. J., Andrew, R. M., Peylin, P., Peters, G. P., Ciais, P., et al. (2021). The consolidated European synthesis of CO₂ emissions and removals for the European Union and United Kingdom: 1990–2018. *Earth System Science Data*, 13(5), 2363–2406. https://doi. org/10.5194/essd-13-2363-2021
- Pisso, I., Sollum, E., Grythe, H., Kristiansen, N. I., Cassiani, M., Eckhardt, S., et al. (2019). The Lagrangian particle dispersion model FLEX-PART version 10.4. Geoscientific Model Development, 12(12), 4955–4997. https://doi.org/10.5194/gmd-12-4955-2019
- Ramonet, M., Ciais, P., Apadula, F., Bartyzel, J., Bastos, A., Bergamaschi, P., et al. (2020). The fingerprint of the summer 2018 drought in Europe on ground-based atmospheric CO₂ measurements. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1810), 20190513. https://doi.org/10.1098/rstb.2019.0513
- Riederer, M., Serafimovich, A., & Foken, T. (2014). Net ecosystem CO₂ exchange measurements by the closed chamber method and the eddy covariance technique and their dependence on atmospheric conditions. *Atmospheric Measurement Techniques*, 7(4), 1057–1064. https://doi. org/10.5194/amt-7-1057-2014
- Rödenbeck, C., Zaehle, S., Keeling, R., & Heimann, M. (2020). The European carbon cycle response to heat and drought as seen from atmospheric CO₂ data for 1999–2018. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1810), 20190506. https://doi.org/10.1098/rstb.2019.0506
- Rodriguez, E., Lefvert, A., Fridahl, M., Grönkvist, S., Haikola, S., & Hansson, A. (2020). Tensions in the energy transition: Swedish and Finnish company perspectives on bioenergy with carbon capture and storage. *Journal of Cleaner Production*, 124527.
- Ruimy, A., Dedieu, G., & Saugier, B. (1996). TURC: A diagnostic model of continental gross primary productivity and net primary productivity. *Global Biogeochemical Cycles*, 10(2), 269–285. https://doi.org/10.1029/96gb00349
- Running, S. W., Baldocchi, D. D., Turner, D. P., Gower, S. T., Bakwin, P. S., & Hibbard, K. A. (1999). A global terrestrial monitoring network integrating tower fluxes, flask sampling, ecosystem modeling and EOS satellite data. *Remote Sensing of Environment*, 70(1), 108–127. https:// doi.org/10.1016/s0034-4257(99)00061-9
- Schaefer, K., Collatz, G. J., Tans, P., Denning, A. S., Baker, I., Berry, J., et al. (2008). Combined simple biosphere/Carnegie-Ames-Stanford approach terrestrial carbon cycle model. *Journal of Geophysical Research: Biogeosciences*, 113(G3). https://doi.org/10.1029/2007jg000603
- Schmidt, M., Reichenau, T. G., Fiener, P., & Schneider, K. (2012). The carbon budget of a winter wheat field: An eddy covariance analysis of seasonal and inter-annual variability. Agricultural and Forest Meteorology, 165, 114–126. https://doi.org/10.1016/j.agrformet.2012.05.012
- Schulze, E., Lloyd, J., Kelliher, F. M., Wirth, C., Rebmann, C., Lühker, B., et al. (1999). Productivity of forests in the Eurosiberian boreal region and their potential to act as a carbon sink—a synthesis. *Global Change Biology*, 5(6), 703–722. https://doi.org/10.1046/j.1365-2486.1999.00266.x
- Serreze, M. C., Walsh, J. E., Chapin, F. S., Osterkamp, T., Dyurgerov, M., Romanovsky, V., et al. (2000). Observational evidence of recent change in the northern high-latitude environment. *Climatic Change*, 46(1–2), 159–207. https://doi.org/10.1023/a:1005504031923
- Siqueira, M. B., Katul, G. G., Sampson, D. A., Stoy, P. C., Juang, J., McCarthy, H. R., & Oren, R. (2006). Multiscale model intercomparisons of CO₂ and H2₀ exchange rates in a maturing southeastern US pine forest. *Global Change Biology*, 12(7), 1189–1207. https://doi.org/10.1111/j.1365-2486.2006.01158.x
- Sitch, S., Friedlingstein, P., Gruber, N., Jones, S. D., Murray-Tortarolo, G., Ahlström, A., et al. (2015). Recent trends and drivers of regional sources and sinks of carbon dioxide. *Biogeosciences*, 12(3), 653–679. https://doi.org/10.5194/bg-12-653-2015
- Smith, B. (2001). LPJ-GUESS An ecosystem modelling framework. (Vol. 12, pp. 22362). Department of Physical Geography and Ecosystems Analysis. INES, Sölvegatan
- Smith, B., Wärlind, D., Arneth, A., Hickler, T., Leadley, P., Siltberg, J., et al. (2014). Implications of incorporating N cycling and N limitations on primary production in an individual-based dynamic vegetation model. *Biogeosciences*, 11(7), 2027–2054. https://doi.org/10.5194/ bg-11-2027-2014
- Smith, N. E., Kooijmans, L. M. J., Koren, G., van Schaik, E., van der Woude, A. M., Wanders, N., et al. (2020). Spring enhancement and summer reduction in carbon uptake during the 2018 drought in northwestern Europe. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1810), 20190509. https://doi.org/10.1098/rstb.2019.0509
- Stoy, P. C., Katul, G. G., Siqueira, M. B. S., Juang, J.-Y., Novick, K. A., Uebelherr, J. M., et al. (2006). An evaluation of models for partitioning eddy covariance-measured net ecosystem exchange into photosynthesis and respiration. *Agricultural and Forest Meteorology*, 141(1), 2–18. https://doi.org/10.1016/j.agrformet.2006.09.001
- Tagesson, T., Schurgers, G., Horion, S., Ciais, P., Tian, F., Brandt, M., et al. (2020). Recent divergence in the contributions of tropical and boreal forests to the terrestrial carbon sink. *Nature Ecology & Evolution*, 4(2), 202–209. https://doi.org/10.1038/s41559-019-1090-0
- Teuling, A. J., Seneviratne, S. I., Stöckli, R., Reichstein, M., Moors, E., Ciais, P., et al. (2010). Contrasting response of European forest and grassland energy exchange to heatwaves. *Nature Geoscience*, 3(10), 722–727. https://doi.org/10.1038/ngeo950
- Thompson, R. L., Broquet, G., Gerbig, C., Koch, T., Lang, M., Monteil, G., et al. (2020). Changes in net ecosystem exchange over Europe during the 2018 drought based on atmospheric observations. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 375(1810), 20190512. https://doi.org/10.1098/rstb.2019.0512
- Tian, H., Xu, X., Liu, M., Ren, W., Zhang, C., Chen, G., et al. (2010). Spatial and temporal patterns of CH₄ and N₂O fluxes in terrestrial ecosystems of North America during 1979-2008: Application of a global biogeochemistry model. *Biogeosciences Discussions*, 7(2). https://doi. org/10.5194/bg-7-2673-2010
- Urban, F., & Nordensvärd, J. (2018). Low carbon energy transitions in the Nordic countries: Evidence from the environmental Kuznets curve. *Energies*, *11*(9), 2209. https://doi.org/10.3390/en11092209
- Van der Velde, I. R., Miller, J. B., Schaefer, K., Van Der Werf, G. R., Krol, M. C., et al. (2014). Terrestrial cycling of 13 CO 2 by photosynthesis, respiration, and biomass burning in SiBCASA. *Biogeosciences*, 11(23). https://doi.org/10.5194/bg-11-6553-2014
- Vincent, W. F. (2020). Arctic climate change: Local impacts, global consequences, and policy implications. In *The Palgrave handbook of Arctic policy and politics* (pp. 507–526). Springer. https://doi.org/10.1007/978-3-030-20557-7_31
- Weedon, G. P., Balsamo, G., Bellouin, N., Gomes, S., Best, M. J., & Viterbo, P. (2014). The WFDEI meteorological forcing data set: WATCH Forcing Data methodology applied to ERA-Interim reanalysis data. *Water Resources Research*, 50(9), 7505–7514. https://doi. org/10.1002/2014wr015638
- WMO. (2019). WMO Statement on the status of the global climate in 2018. World Meteorological Organization.
- Wutzler, T., Reichstein, M., Moffat, A. M., Menzer, O., Migliavacca, M., Sickel, K., et al. (2018). Post processing of (half-) hourly eddycovariance measurements. *R Package Version*, 1(5).



- Xu, B., Arain, M. A., Black, T. A., Law, B. E., Pastorello, G. Z., & Chu, H. (2020). Seasonal variability of forest sensitivity to heat and drought stresses: A synthesis based on carbon fluxes from North American forest ecosystems. *Global Change Biology*, 26(2), 901–918. https://doi.org/10.1111/gcb.14843
- Xylia, M., & Silveira, S. (2016). Leading or lagging in the EU? Sweden's progress towards energy efficiency targets for 2020. Energy strategy Reviews.